A Review of Energy Efficiency Evaluation Technologies in Cloud Data Centers

Saiqin Long, Yuan Li, Jinna Huang, Zhetao Li, Yanchun Li

PII:	\$0378-7788(22)00019-6
DOI:	https://doi.org/10.1016/j.enbuild.2022.111848
Reference:	ENB 111848
To appear in:	Energy & Buildings
Received Date:	2 November 2021
Accepted Date:	9 January 2022



Please cite this article as: S. Long, Y. Li, J. Huang, Z. Li, Y. Li, A Review of Energy Efficiency Evaluation Technologies in Cloud Data Centers, *Energy & Buildings* (2022), doi: https://doi.org/10.1016/j.enbuild. 2022.111848

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2022 Elsevier B.V. All rights reserved.

Saiqin Long^{*a,b,c*}, Yuan Li^{*c*}, Jinna Huang^{*c*}, Zhetao Li^{*a,b,c,**} and Yanchun Li^{*a,b,c*}

^a The Key Laboratory of Hunan Province for Internet of Things and Information Security, Xiangtan 411105, China ^b Hunan International Scientific and Technological Cooperation Base of Intelligent network Xiangtan University, Xiangtan 411105, China ^c The School of Computer Science, Xiangtan University, Xiangtan 411105, China

ARTICLE INFO

Keywords: data center energy efficiency energy consumption evaluation metrics

ABSTRACT

The energy consumption by data centers is expanding in tandem with the rapid rise of the digital economy. Data centers, as high-energy-consumption organizations, have garnered extensive attention from society in order to accomplish energy conservation and emission reduction. As a result, improving the energy efficiency of cloud data centers has become a major topic of research. Researchers are working hard to develop practical energy efficiency evaluation methodologies and metrics in order to attain this goal. This article summarizes data center energy efficiency evaluation methods, classifies existing energy efficiency evaluation metrics, examines the current state and challenges of data center energy efficiency evaluation, and makes recommendations for improving energy efficiency evaluation strategies. We give data center researchers a better grasp of energy efficiency evaluation and encourage them to combine theory and practice in energy efficiency evaluation and encourage them to combine theory and practice. This is a critical step in the quest for the most advanced green technology, as well as a significant step toward reaching sustainable development goals.

1. Introduction

The world as a whole is embarking on a major revolution from the actual to the digital economy. The fundamental support for smart city management and industrial digitization is data, which is the primary production component of the digital economy. With the commercialization of 5G and the rise of artificial intelligence, more large-scale data will be generated. Data center capacity is quickly rising over the world, and this is especially true in China [1]. According to statistics from [2], the main data industry will account for 15% of total economic volume by 2030, and China's total data would exceed 4YB, accounting for 30% of world total data. According to Canalys data, service spending on worldwide cloud infrastructure climbed by 35% to \$41.8 billion in the first quarter of 2021. This is an increase of nearly 11 billion dollars over the first quarter of 2020 [3].

However, as data center sizes grew, so did their power consumption, which soared at an alarming rate every year. The Amsterdam metropolitan government in the Netherlands banned the construction of new data centers because the rapid growth in the number of data centers in metropolitan areas has resulted in insufficient space and significant demand on the power grid [4]. Each year, data centers in the United States require more than 90 billion kilowatt-hours of electricity, necessitating the construction of 34 largescale (500 MW) coal-fired power plants. Global data center energy consumption is over 416 terawatts (around 3% of total

Saiqin LONG et al.: Preprint submitted to Elsevier

electricity), and it doubles every four years [5]. Some studies have confirmed the consensus [6, 7, 8] that data center electricity demand will rise from 200 billion kWh in 2010 to 2,000 to 3,000 billion kWh by 2030. Meanwhile, the rate of increase in the global carbon footprint is alarming. Because the data center is an increasing power-intensive industry and a carbon emission giant in the Internet era, it is needless to explain the importance of energy efficiency research. Data center energy efficiency evaluation has piqued the interest of data center operators, decision-makers, and researchers due to the high energy consumption behavior of data centers. It is vital to understand the energy consumption composition and hierarchical structure of the data center before undertaking energy efficiency evaluation research. Figure 1 depicts the power consumption of various components in data centers in the United States, which shows that servers account for 43% of the total, followed by the storage driver. Despite the fact that network devices make up only 3% of the total, they must not be overlooked.

IT equipment and physical infrastructure make up the majority of data centers. The physical infrastructure primarily consists of power supply and temperature control systems. Figure 2 depicts the relationship model. All IT equipment must be operated with extraordinary dependability at all times. The temperature control equipment mostly consists of air conditioning, heating, ventilation, and other similar items. The temperature control is primarily employed to avoid overheating caused by the enormous quantity of heat created by IT equipment during operation, which also uses a significant amount of additional electricity. The power supply facilities in data centers mainly include UPS, PDU transformers, connectivity(cabling) and backup generators, etc. Servers, storage, and networks are the most common

^{*}Corresponding author

xxgcxyxtu@sina.com (S. Long); li_yuan28@163.com (Y. Li); hjn2018@126.com (J. Huang); liztchina@hotmail.com (Z. Li); ycli@xtu.edu.cn (Y. Li)

ORCID(s):

A Review of Energy Efficiency Evaluation Technologies in Cloud Data Centers



Figure 1: Fraction of electricity used by US data centers in 2014 [1, 9].





types of IT equipment, according to [10]. Because of the various application scenarios in different data centers, the classification of IT equipment is slightly different. According to several studies [11, 12], IT equipment is primarily made up of servers arranged in racks. Application servers, storage servers, and network devices such as switches, routers, and other network devices are among the servers [13]. Unlike other articles, Dayarathna et al. [14] advocated that IT equipment should be classified based on the kind of hardware components, taking into account the primary sources of energy consumption. IT systems should be split between hardware and software, according to several reviews [14, 15]. The hardware is divided into three parts: Server Tier, Storage

Tier, and Network Tier, which correspond to the three main components: servers, storage drives, and networks.

Data center operators spend a lot of money on electricity. More effective measures are needed to establish a healthy link between human development and environmental preservation, given the high loss rate and high energy consumption of the data center industry. Effective energy efficiency evaluation methods and standards are required for the data center industry to achieve its objectives. Researchers in data centers can come up with a variety of strategies to lower the data center's energy usage [16, 17, 18]. When energy consumption decreases, how do we know whether the reduction in energy consumption will affect the performance of

A Keview of Energy Efficiency Evaluation Technologies in Cloud Data Centers

Table 1

Summary of	the	review	articles on	data	center	energy	efficiency
------------	-----	--------	-------------	------	--------	--------	------------

Classification	Theme	Years	Ref
Energy efficiency	The State of Practice of Power Measurement and Energy Efficiency Technology in the Data Center Industry	2016	[21]
Thermal management	Different levels of thermal performance evaluation metrics for data centers	2020	[22]
Energy efficiency	Energy efficiency and demand response of small and medium data centers	2018	[23]
Thermal management	Thermal management and evaluation technology for data centers and related cooling systems	2018	[24]
Metric performance	Green performance metrics for data center environmental performance	2013	[9]
Energy efficiency	The main technology of cloud computing energy saving	2016	[25]
Thermal management	The impact of airflow on the thermal environment and energy efficiency of data centers	2019	[26]
Thermal management	Data center air conditioning energy performance standards	2017	[27]
Metric performance	Defects and improvement directions of PUE metrics	2016	[28]
Energy consumption of components	Power consumption modeling and optimized control strategy of cooling system	2021	[29]
Energy consumption of components	Energy consumption model and modeling method of cloud server	2020	[30]
Energy consumption of components	The impact of mechanisms used at the hardware component level on data center power consumption	2021	[31]

the data center? Energy efficiency evaluation is a method to effectively measure energy utilization and provides guidance for energy conservation in data centers [19, 20]. Therefore, the energy efficiency evaluation of data centers is considered to be one of the most concerned research points for cloud operators.

Table 1 lists survey articles about data center energy efficiency so far. It can be found that most of the reviews focus on energy efficiency, metric performance, thermal management and component energy consumption. The energy efficiency reviews focused on analyzing the energy distribution of the data center and energy efficiency improvement technologies. For example, virtualization technology and energy-aware scheduling. However, there are few introductions to evaluation metrics and evaluation methods. The reviews of metric performance mainly focuses on energy efficiency evaluation metrics, and less summary of energy efficiency evaluation methods. We summarized the metrics in three granularities, and then analyzed the advantages and disadvantages of commonly used metrics, especially PUE. The reviews of thermal management and component energy consumption mainly focused on part of the energy efficiency of the data center. This review is dedicated to energy efficiency evaluation techniques for data centers. We summarized the methods and metrics of data center energy efficiency evaluation, and discussed the current status, challenges and recommendations of energy efficiency evaluation. The main contributions are as follows:

1) We summarized the three approaches for evaluating data center energy efficiency and compared their benefits and drawbacks. For energy efficiency evaluation, a multimetric fusion evaluation was recommended. 2) We analyzed the current state of energy efficiency evaluation technology in industry and academia, and gave recommendations to data center operators, decision-makers, and researchers.

3) We summarized the data center energy efficiency evaluation metrics, as well as their benefits and drawbacks, so that users could select the most relevant metrics. Then we looked at the problems in the existing energy evaluation metrics and made recommendations for improving and inventing new PUE metrics.

The remainder of this paper is organized as follows. Section 2 describes the three approaches for evaluating data center energy efficiency and compares and contrasts their benefits and drawbacks. Section 3 summarizes the energy efficiency evaluation metrics of data centers and analyzes the benefits and drawbacks of commonly used metrics, particularly PUE. Section 4 analyzes the current state and challenges of energy efficiency evaluation, and provides a reference for the next step of research in the field. Section 5 put forward suggestions to improve energy efficiency evaluation technology in order to cope with the development of data centers. Section 6 summarizes the article and gives recommendations for data center operators, decision-makers, and researchers in terms of future directions.

2. Energy efficiency evaluation methods

The present physical data center and the data center to be developed must both be considered when evaluating data center energy efficiency. The methods for evaluating energy efficiency are summarized in this section. Approaches for evaluating data center energy efficiency can be split into

A Review of Energy Efficiency Evaluation Technologies in Cloud Data Centers

Measurement method of energy efficiency metrics			
Measurement method	Study focus	Ref	
Linear function	Measurement of power consumption of multi-component based on cloud server in heterogeneous cloud environment	[36]	
Machine learning	Prediction of key performance metrics related to data center power efficiency	[37]	
Machine learning	Performance metric of HPC data center's hot water cooling circuit	[38]	
Expert system based on trust rules	Prediction of data center power utilization efficiency (PUE)	[39]	
Linear function	A framework for rapid estimation of application energy consumption in data centers	[40]	

Table 2

three categories: measurement-based methods, simulation-
based methods, and analytical modeling-based methods [32,
33]. The measurement-based evaluation is the most accurate
of the three. The accuracy of the simulation-based evaluation
is the lowest.

2.1. Measurement-based evaluation

At present, energy efficiency evaluation is mainly completed by a combination of short-term on-site measurements and engineering calculations. Measurement-based evaluation is the most intuitive, basic and accurate method. However, the measurement process requires a lot of manpower and financial resources. To measure metrics linked to data center energy efficiency, this method normally necessitates the use of specialized equipment. For example, researchers frequently utilize a Power Distribution Unit (PDU) to measure the power consumption of IT devices. Every measuring instrument has a certain degree of inherent error. Some researchers have attempted to develop novel measurement methods in order to increase the accuracy of measurement equipment. Some measurement methods are shown in Table 2. The accuracy of measurement evaluation depends not only on the accuracy of the measurement equipment, but also on the evaluation metrics. PUE was proposed as an energy efficiency indicator for data centers by the US Green Grid Organization in 2007[34]. PUE is frequently utilized in the data center industry as a simple and easy-to-use evaluation metric [35]. However, because to the limits of PUE, data centers may only be evaluated in this manner in a rough sense. Furthermore, measurement-based evaluation methods are only applicable to systems that are already in place. We'll go over energy efficiency evaluation metrics in more detail in Section 3. In the realm of data centers, there is currently no approved measurement method. Central and Eastern European countries have developed a set of energy efficiency evaluation guidelines based on actual measurements¹, but only focus on water resource utilization and carbon emissions, not data center energy efficiency.

2.2. Simulation-based evaluation

It is difficult to evaluate the solution in a real setting when dealing with large-scale computing systems, which is a significant drawback in evaluating data center energy

efficiency. To circumvent this constraint, simulation tools have rapidly become the most common way of preliminary testing [65, 66]. Simulation tools offer a versatile technique to assess data center energy efficiency. It only takes a quick assessment of energy efficiency on a simulated cloud computing system. It can also assess not only existing systems but also those that have yet to be developed. CloudSim [41] and GreenCloud [42] are two of the most used simulation tools. CloudSim was developed by the Grid Lab and the Gridbus project of the University of Melbourne, Australia. In order to make up for the shortcomings of CloudSim, most of the newly developed simulation tools have been extended on the basis of CloudSim to meet the special needs of data centers [67, 44, 48, 43]. Table 3 shows the existing simulation tools. Simulation tools have obvious flaws. The use of simulation tools to evaluate energy efficiency, according to Kafhali et al. [33], necessitates independent numerous runs of simulation models to preserve the effects of varied input settings, which is time-consuming and can lead to incorrect results. Furthermore, there are some flaws in the simulation platform's energy consumption model. As a result, using simulation methodologies to assess the energy efficiency of cloud data centers is not recommended. However, in certain instances, it can be used as a preliminary test.

2.3. Analytical modeling-based evaluation

Analytical modeling-based evaluation employs mathematical techniques to create appropriate models that abstract the system (whether it exists or not) and define the link between performance, energy consumption, and load. It significantly decreases the burden associated with evaluating energy efficiency. The determination of model parameters is based on the system's measurement findings or a parameter guess that differs from the real metrics. The accuracy of performance and energy consumption metrics depends on the accuracy of the model. Simplifications and assumptions are common in analytical modeling-based evaluation methodologies. Faced with various behaviors of complex systems, it lacks the ability to describe the details of the system. One strategy to increase the model's performance is to employ more characteristic variables to represent the system's specifics. However, simply adding feature variables to a model does not guarantee that it will improve. Because some factors will have a detrimental impact on the model's performance. More researchers are turning to highly reliable

https://www:cee1:org/LearningAboutEvaluation

Table 3

Simulation analysis of performance and energy consumption

Simulation tools	Study focus	Ref
CloudSim	Performance of cloud resource allocation strategies, application workload models, and resource performance models	[41]
GreenCloud	Work load distribution, component energy consumption in the data center, and packet-level communication mode	[42]
ThermoSim	Thermal perception of cloud data centers nodes based on deep learning	[43]
AutoScaleSim	Supports automatic extension of web applications in a customizable, extensible and scalable manner in a cloud environment	[44]
SinergyCloud	Evaluate hybrid cloud scenarios with fine abstract granularity, including energy consumption, workflow completion time, task completion time, and virtual machine migration	[45]
IoTSim-Osmosis	In a heterogeneous edge cloud environment, support penetration computing for complex IoT applications	[46]
MT-EA4Cloud	Check the correctness of the cloud system from the perspective of energy-aware and optimize its energy consumption	[47]
MultiRECloudSim	Extend CloudSim to implement multi-resource scheduling and power consumption modeling	[48]

Table 4

Modeling high availability to analyze performance and energy

Modeling method	Study focus	Ref
Stochastic Reward Net(SRNs)	Investigate data center performance and QoS in IaaS Clouds	[49]
Queue theory Markov chain	Analyze performance and energy consumption in cloud data centers	[33]
Stochastic model based on Queueing theory	Stochastic modeling of dynamic right-sizing for data center efficiency	[50]
Bayesian networks	Allocation of cloud resources	[51]
Probability distribution	Model the accumulated downtime	[52]
Statistical distribution	Model a service-based market	[53]
K-means and TOPSIS	Building energy performance evaluation and ranking	[54]
Emergy analysis	Energy consumption of data center cooling system	[55]
Machine learning	Energy efficiency based on changes in behavior	[56]
Machine learning	Assess the load and energy consumption of the data center	[57]
K-type clustering algorithm	Develop a dynamic room-level energy benchmark test method	[58]
Grey model and risk evaluation model	Energy security in China	[59]
Mixed multi-criteria decision	The economic-environment-energy (3E) performance of the CCHP-MG	[60]
	system	
Data analysis	Data center dedicated cooling system	[61]
Neural network and fuzzy comprehensive		
evaluation method	Carbon efficiency performance evaluation	[62]
Data mining	Anomaly detection and dynamic energy performance evaluation of	[63]
	HVAC systems	[03]
Energy decomposition	Energy performance of district heating substations	[64]

approaches to develop cloud models and examine the tradeoffs between performance and energy efficiency [2, 55, 57, 61]. Table 4 shows some representative analytical modelingbased evaluation methodologies. Machine learning, particularly deep learning, can abstract complex systems better than typical methods. As a result, academics are attempting to simulate using machine learning [56, 57, 62].

3. Energy efficiency evaluation metrics

Energy efficiency metrics are the measurement standard of energy efficiency evaluation, which are beneficial for operators to better understand potential inefficiencies and improve their performance by analyzing core parameters [68]. It's useful to promote industry innovation and environmental sustainability goals for researchers and practitioners in data centers. Therefore, research on energy efficiency metrics is the focus of energy efficiency evaluation. In 2017, Reddy et al.[11] analyzed 132 sustainability-related metrics of data centers. From the perspective of data center operations, energy efficiency metrics mainly included cooling, performance, sustainability, air management, security, financial impact (data center energy changes, carbon, oil price changes), etc. At the same time, they analyzed the relationship between metrics. The summary contained the

Journal Pre-proofs A Review of Energy Efficiency Evaluation Technologies in Cloud Data Centers

Tabl	е	5					
Coai	se	e-gr	anu	larit	tv n	netr	ics

Metric	Name	Description	Ref
PUE	Power Usage Efficiency	Power Usage Efficiency is the current main energy efficiency metric of data centers	[35]
DCiE	Data Center Infrastructure Efficiency	A performance improvement metric used to calculate the energy efficiency of a data center	[72]
DCeP	Data Center Energy Productivity	It incorporates both the infrastructure and the IT equipment when evaluating data center energy efficiency	[73]
CADE	Corporate Average Data Center Efficiency	Performance metric that evaluate the energy efficiency of a data center can be used to measure the performance of IT equipment and facilities, and can be compared with the performance of other data centers	[74]
TUE	Total-Power Usage Effectiveness	TUE is defined as the ratio of total energy consumption to the specific energy used by the computing components, which makes up for the shortcomings of PUE	[75]
DCPE	Data Center Performance Efficiency	DCPE is defined as the ratio of useful work to the total facility power	[76]

most common energy efficiency metrics except IPA(the Infrastructure Power Adaptability), PVar(the Power Variability) [69], and OEF(On-site Energy Function), OEM(Onsite Energy Matching) [70]. The work in [15] and [11] has comprehensively collected and analyzed many data center energy efficiency metrics, which can be used as a data center metric manual.

3.1. Classification of metrics

The structure of a modern cloud data center is quite complex and involves many IT equipments and infrastructures. With such a complicated structure, defining energy efficiency evaluation metrics that are adaptable is quite difficult. Besides, the measurability of metrics in practice must be considered. The existing energy efficiency evaluation metrics of data centers can be divided into the following three categories from granularity: coarse-granularity metrics, medium-granularity metrics, and fine-granularity metrics [71].

3.1.1. Coarse-granularity metrics

Many metrics have been proposed at the level of the entire data center by researchers. These metrics are a little rought, but they're easy to come by. PUE [89] is a typical metric that assesses data center energy efficiency as a ratio of total data center energy consumption to energy consumption of IT equipment, which is relatively concise compared to other metrics. The representative coarse-granularity energy efficiency evaluation metrics are summarized in Table 5. These metrics are popular because they employ only one metric to evaluate the data center's total energy efficiency, which eliminates the need for several details in the data center. However, because energy efficiency details of data center sub-components are lacking, the evaluation results cannot be used to advise data center energy savings.

3.1.2. Medium-granularity metrics

Researchers have proposed a significant variety of mediumgranularity metrics for data center components. The energy consumption behavior of data center subcomponents is ignored by coarse-granularity metrics. To compensate for the lack of coarse-granularity metrics, medium-granularity metrics give energy efficiency information for specific equipment, infrastructure, and green energy in the data center. Table 6 summarizes the representative medium-granularity energy efficiency metrics by IT equipment, cooling equipment, and green energy. Medium-granularity metrics can evaluate the energy efficiency of data centers more accurately and extensively than coarse-granularity metrics due to their diversity and pertinence.

3.1.3. Fine-granularity metrics

The fine-granularity metrics² provide detailed performance information for specific components in data centers. These components are IT equipment responsible for the normal operation of data centers, such as CPU, disk, memory, network, and storage. These performance metrics cannot directly represent data center energy efficiency, but they may be evaluated by looking at changes in performance and energy consumption, which can be seen through the performance/watt rate. The fine-granularity metric of energy efficiency evaluation is the most significant thing for cloud operators. Users are mainly interested with performance metrics, and operators seek to deliver improved services while decreasing energy use to the greatest extent possible. Coarse-grained and medium-grained energy efficiency evaluation methods often ignore the needs of customers. The representative data center fine-granularity metrics is summarized in Table 7.

²https://docs.vmware.com/

Table 6	
---------	--

Medium-granularity metrics

Туре	Metric	Full Name	Description	Ref
	ITEE	IT Equipment Energy	Measurement of IT Equipment Energy.	[77]
	SCE	Server Compute Efficiency	Determining whether the server is performing tasks and measuring the proportion of useful work.	[78]
	DWPE	Data center Workload Power Efficiency	Energy efficiency ratio of a specific workload and the overhead for operating a given system in a certain data center.	[79]
	EPI	Energy Proportionality Index	It depicts the proportional relationship between energy consumption and load.	[80]
IT Equipment	ITEU	IT Equipment Utilization	Characterizing the operating efficiency of IT equipment in data centers.	[77]
	TGI	The Green Index	Evaluation of the energy efficiency of severs based on various existing benchmark tests, Weights, such as time, energy and consumed power.	[81]
	CNEE	Communication Network En- ergy Efficiency	Ratio of power consumed by network equipment and effec- tive network throughput capacity, which is used to evaluate the energy of delivering a single message.	[82]
	NPUE	Network Power Usage Effec- tiveness	Ratio of total power consumed by IT equipment and power consumed by network equipment.	[82]
	COP	Coefficient of Performance Ensemble	Ratio of total power consumption of HVAC equipment to power consumption of IT equipment.	[83]
	RER	Renewable Energy Ratio	Measurement of the utilization of renewable energy in data centers.	[84]
Cooling Equipment	DCCSE	Data Center Cooling System Efficiency	Evaluation of the efficiency of the HVAC system in terms of power used per unit of cooling output.	[85]
	MLC	Mechanical Load Components	Ratio of total power consumption of HVAC equipment to power consumption of IT equipment per year.	[86]
	CLF	Cooling Load Factor	Ratio of cooling equipment power to IT equipment power.	[83]
	WUE	Water Usage Effectiveness	Evaluation of water usage efficiency in data centers.	[87]
	CUE	Carbon Usage Effectiveness	Evaluation of carbon usage efficiency in data centers.	[88]
Green	ERE	Energy Reuse Effectiveness	Measurement of the reused energy. Ratio of energy other	[84]
Energy			than reused energy to IT equipment energy.	
	GEC	Green Energy Coefficient	Ratio of renewable energy used to total power consumption.	[77]

Table 7

Fine-granularity metric

Туре	Name	Description
CPU	CPU Utilization	Proportion of CPU working time in total time
	CPU Workload Balance Factor	Measuring the ability to allocate tasks across multiple operating units
	Waiting Time	Waiting time of each process in the ready queue
Diek	Lisaga Pata	Sum of the average read data and written data of all disk instances of
DISK	Usage Rate	hosts or virtual machines
	Commands Per Second	Average number of commands issued within a time interval per second
Memory	Memory Usage	Memory usage of hosts
Network	Throughput	Maximum data rate that devices can accept and forward
	Data Transmission Speed	Average amount of data transferred per second
Storage	Storage Usage Average	Total throughput rate

3.2. Advantages and disadvantages of metrics

Coarse-granularity metrics focus on the overall energy consumption behavior of the data center while overlooking specific nuances. Medium-granularity metrics are used to supplement coarse-granularity metrics and focus on data center sub-component energy consumption behavior. The majority of fine-granularity metrics are tied to IT equipment and cannot be used to determine energy efficiency directly. For data center researchers and practitioners, there are currently hundreds of data center energy efficiency metrics. In order to select appropriate metrics for energy consumption evaluation, we also need to understand the advantages and disadvantages of different metrics.

Table 8

Advantages and d	lisadvantages	of data	center	metrics
------------------	---------------	---------	--------	---------

Metric	Advantages	Disadvantages	
PUE	Evaluation metrics recognized by the industry, easy to measure	Too simple, without considering the details	
DCiE	The inverse value of PUE , easy to measure	Unable to capture small changes	
DCeP	Closest to the definition of energy efficiency	Difficult to measure	
CADE	Can be used for data center performance comparison	Difficult to measure and complicated to calculate	
TUE	Solving the problem that PUE increases when hardware energy consumption is reduced due to the development of technology	Hardware energy consumption is difficult to measure and can only be predicted	
DCPE	Can measure data center performance efficiency	Difficult to define useful work and measure	
WUE	Increase the evaluation of water use to make up for the lack of PUE	The calculation process is cumbersome	
CUE	Pay attention to the sustainability of the data center's carbon footprint	Carbon dioxide emissions are difficult to accurately estimate	
EPI	Pay attention to the proportion of energy in the data center	Difficult to use load to represent hardware utilization	

3.2.1. Advantages and disadvantages of important metrics

Fine-granularity metrics cannot directly evaluate the energy efficiency of data centers. They are not commonly used. Currently, the industry mostly uses coarse-granularity metrics or medium-granularity metrics to evaluate data centers energy efficiency. Table 8 summarizes the advantages and disadvantages of some important metrics. PUE is one of the most extensively utilized metrics in the industry. This metric will be discussed further below.

3.2.2. Advantages and disadvantages of PUE

At present, data centers still use PUE as the main measurement standard for energy efficiency evaluation. It offers a number of advantages, including ease of measurement and promotion, and its calculation method is not limited by servers, storage capacity, or data center construction. However, as the data center industry has grown, so have the demands for great energy efficiency and long-term sustainability. PUE has gradually revealed some flaws. There are many specific studies aimed at improving PUE [90, 91, 92]. We summarize the disadvantages of PUE to help related researchers improve it.

The disadvantages of PUE can be summarized as follows:

- (1) PUE merely summarizes the overall energy efficiency of data centers. It lacks practical guidance due to a lack of detail. Based on the evaluation results of PUE alone, we have no way to know how to improve the energy efficiency of data centers.
- (2) According to the definition of PUE, we fail to know the energy efficiency of IT equipment from the metric [12]. While IT equipment is exactly the backbone of data centers. When we try to improve IT equipment for efficiency, we don't know where to start. It's unknown whether all the electricity used by IT devices yields beneficial output. In terms of the value of PUE, the productivity of the data center is a mystery.

- (3) In order to meet the requirements of sustainable energy development. The energy efficiency evaluation of the modern data center industry not only evaluates the energy efficiency utilization of data center IT equipment, but also reflects multiple aspects such as water utilization, carbon utilization, and renewable energy utilization. It is impractical to try to use one indicator to cover all aspects of data center energy efficiency. What is important is that researchers need to pay attention to the impact of energy efficiency on green resources. Therefore, we need to find better energy efficiency evaluation methods.
- (4)With the advent of virtualization technology, the energy efficiency of data centers has improved. However, if the scale of infrastructure is not compatible with the reduction of the overall load, the efficiency of infrastructure will be reduced. In this case, the result of PUE will be worse, which shows that PUE has great limitations as a measurement standard [93]. Van et al. [70] summarized the impact of IT load on PUE. First, the value of PUE was better when the IT load was relatively high [94]. Second, when more efficient IT devices were installed, the drop in average IT load will cause a drop in PUE [95]. While PUE is only used to measure the efficiency of the physical infrastructure architecture in data centers, rather than IT computing power efficiency. It's a failure to reflect obvious environmental benefits after reducing energy consumption, which is a serious defect as a widely used metric of energy efficiency.
- (5) PUE considers the energy used for cooling regardless of temperature. As a result, it is theoretically impossible to compare different data centers [90]. To reduce the cost of thermal energy, Microsoft in the United States chose to sink the data center to the ocean floor. And China's Tencent reduced cooling costs by digging holes. It can be seen that the effect of temperature on the energy efficiency of data centers cannot be ignored because of the climate difference in different regions. And PUE's

original purpose was never used for comparison, but it is currently used for this purpose [11]. Because much more energy burden is invested to cool the data center in tropical regions than that in freezing regions, it is unreasonable to use PUE to perform a unified energy efficiency evaluation comparison under huge temperature differences.

4. Status and challenges of energy efficiency evaluation

4.1. Status of energy efficiency evaluation

In order to improve the energy efficiency of data centers, the data center industry and researchers have developed a large number of energy efficiency evaluation techniques. The efforts of industry and researchers have had a positive impact on data center energy efficiency. To help the industry and researchers further improve the energy efficiency of data centers, we analyzed the current status of energy efficiency evaluation in industry and academia.

4.1.1. Data center industry

The industry has been working on energy efficiency evaluation metrics for many years. The Green Grid Organization proposed the principles of measuring DCeP, PUE, GEC, ERF, and CUE [96]. ISO and IEC have established a number of standards for energy efficiency evaluation criteria in recent years. In 2016, they formulated standards for key performance metrics for data centers, and proposed that key performance metrics must cover the effective use of resources and reduce carbon dioxide emissions [97, 98]. In the latest research, they have developed data center server energy efficiency metrics (SEEM) [99] and application platform energy effectiveness (APEE) [100] standards to improve key performance metrics. The CEN/CENELEC/ETSI Green Data Center Coordination Group defined data center design standards in their most recent report, stating that the industry must set greater targets for energy management and environmental feasibility in the future [101]. For example, the influence of air pollution, condensation, and toxic negative products on data centers. Standards for evaluating data center energy efficiency have always been developed by the industry, not by academics.

With the joint efforts of many stakeholders, relevant data shows that the PUE value has been declining year by year [102]. With the development of science and technology, it is a question of whether the past energy efficiency evaluation technology is suitable for new data centers. Moreover, some facts indicate that the energy consumption of data centers will still increase substantially [103]. With the increase in the number of computing instances, the proportion of servers in ultra-large-scale data centers continues to increase [9, 104, 105]. The energy efficiency benefits of servers are not enough. As a result, the industry has a long way to go in terms of improving data center energy efficiency.

4.1.2. Energy efficiency research

Hundreds of metrics have been devised by researchers to evaluate data centers, however they are rarely used. In the early period, an unprecedented survey on the application of data center efficiency metrics was described by Procaccianti G et al. [106]. The results showed that less than a third of the respondents (461 out of 1523 respondents) knew about data center efficiency metrics. Most employees in the data center industry lacked awareness of evaluation metrics. **Researchers** did another similar poll four years later [15], and many data center personnel still knew very little about energy efficiency evaluation metrics.

As mentioned above, the research results are out of touch with industrial production. The lack of understanding of energy efficiency metrics makes it difficult to keep up with the pace of technological change. Only when energy efficiency metrics are properly pushed is an academic study on energy efficiency worthwhile.

Furthermore, rather than the thorny challenges encountered in the data center industry, the research of energy efficiency evaluation should be based on actual requirements. The most effective technique to determine the most appropriate assessment method is to do a specific study for each problem. However, it is regretful that much energy efficiency evaluation work is restricted to academia, leaving it utterly disconnected from decision-makers and businesses.

4.1.3. Misuse of metrics

One of the great problems on data center energy efficiency evaluation in the twenty-first century is the misuse of energy efficiency metrics [107]. For example, PUE plays a significant role in the energy efficiency evaluation by environmental protection personnel. When taking climatic factors into account, if the local perennial temperature is high, this means that more energy needs to be consumed for cooling. In this case, the PUE evaluated will conclude that the data center should not be established. However, because the PUE was designed to quantify the energy efficiency of data center infrastructure, the results of measuring data center energy efficiency using the PUE after server integration are unreliable. It can be seen that different application scenarios require different energy efficiency metrics in data centers to evaluate energy efficiency. Misuse of metrics may adversely affect the operation management of data centers and the research of scholars.

Green Grid clearly stated in the PUE white paper that it is not recommended to use PUE to compare different data centers [35]. However, the PUE has become a unified standard in the industry, and people have forgotten whether it can be used to compare different data centers. The internal structure of different data centers is different, and the corresponding measurement methods are also different. The abuse of metrics has brought disasters to the data center energy efficiency evaluation.

4.2. Challenges of energy efficiency evaluation

At present, the energy efficiency evaluation of data centers has not yet formed a systematic theory, and there is no

set of universal standards. Most of the metrics in Section 3 need to be measured from the actual data center, but due to the difference in the structure of the data center, there are many limitations in energy efficiency evaluation.

Most of the data center energy efficiency metrics are aimed at a local system or a specific target, which lacks practicality. The shortcomings of this data center energy efficiency rating approach can be attributed to a variety of factors. For starters, establishing universal energy efficiency evaluation standards is difficult due to the diversity and complexity of data center structures. In data center construction, many elements must be addressed, including geographic location, floor area, local climate, construction scale, and interior design. Second, the technology for measuring data center energy efficiency has yet to be broken. Third, there is no worldwide body that plans data centers and establishes common standards for data centers in different nations.

Most of the current energy efficiency evaluation standards are mainly developed for existing physical data centers. A large number of researchers expect to have an omnipotent data center energy efficiency evaluation metric or method, which can reflect both the energy efficiency of infrastructure and IT equipment. However, It is really difficult to have both. These two parts are important components of data centers. They are not mutually restricting nor antagonistic, despite the fact that they are not fully separated. The energy efficiency of IT equipment will not be sacrificed for cooling, and vice versa. The relationship between the two is that cooling equipment is the same as the various facilities performing duties in the data center, as auxiliary equipment for IT equipment. So, the energy evaluation metric is specifically aimed at IT equipment shall be designated for the desire to reflect the energy efficiency of IT equipment.

All in all, it's impossible to expect that an energy efficiency metric of data centers can reflect the energy efficiency of IT equipment and carbon utilization rate, water utilization rate, heat recovery, renewable energy utilization rate, cooling efficiency, safety, etc. A metric can be regarded as a specific metric for a specific problem. For example, a major issue requires large and general metrics, while a subtle issue requires refined metrics. Regarding the energy efficiency evaluation of a modern data center, many energy efficiency issues need to be considered, such as improving the utilization rate of renewable energy, improving cooling efficiency, etc. We can specify a metric system to evaluate the energy efficiency of data centers, instead of trying to use an evaluation metric replacing PUE to apply contents in many aspects like IT equipment utilization rate, carbon utilization rate, water utilization rate, renewable energy utilization rate. Fernando et. al [21] made a good demonstration, who developed a set of metrics to evaluate the overall energy efficiency of data centers, calculated metrics by using accumulated energy requirement as a resource metric and then analyzed metrics through the life cycle of products.

5. Proposals for energy efficiency evaluation

Energy efficiency metrics are the flesh and blood of energy efficiency evaluation. In response to the challenge of energy efficiency evaluation, we propose several methods to improve the existing evaluation methods of data centers. We advocate initially enhancing the PUE performance because it is now the most essential energy efficiency evaluation method and the ancestor of energy efficiency evaluation. At the same time, we pointed out that new metrics should be devised to address the industry's shifting issues. Finally, to completely capture the specifics of data center energy efficiency, we offer a multi-metric fusion method.

5.1. Improvements of the PUE

Although the PUE has many defects (Section 3.2.2), it has been greatly promoted due to the simplicity of the model. The PUE is still a retrospective metric, in essence, based on the values measured by data center operators, and it cannot explain the potential factors that determine the PUE value (such as installed equipment and operator decisions, etc) [108]. Some researchers tried to improve PUE to promote cloud operators and decision-makers to better make data center strategies. Voort [109] used supplementary metrics to find the potential of energy efficiency metrics outside the scope of PUE. In order to better evaluate energy efficiency, the GUE (grid utilization efficiency) was introduced. But instead of verifying the general model against actual PUE data, it is a simplified calculation to compare the theoretical PUE reduction under different natural cooling assumptions. Google [110] has tried to apply machine learning to data center optimization. This model can predict PUE through learning from actual operational data and achieve a 0.4% error. This method needs a huge number of non-public data sets, making replication and promotion challenging.

Nuoa Lei et al. [108] conducted the most comprehensive study, developing a statistical framework for predicting and analyzing data center power usage efficiency (PUE) based on the thermodynamic PUE model. Climate variables and energy system parameters are input into the PUE prediction model. The PUE value is predicted under uncertainty and verified using the reported values of real data centers. These studies all have considered the temperature factors which have a significant impact on data centers and minimized the impact of climate differences and seasonal changes on the accuracy of the energy efficiency evaluation results. In fact, the PUE energy efficiency model based on simple measurements is outdated. It is necessary to consider the factors of differences in temperature and system parameters in data centers to improve the PUE to meet the rapid development of data centers. Additionally, it is recommended that different data centers should be graded, and PUE can be compared among different data centers after being rated as different levels of size.

5.2. Development of new energy efficiency metrics Organization that develops data center evaluation metrics: standardization bodies, industry groups, professional

bodies, and local, regional, and international government agencies. It is strongly believed that all the proposed metrics are usually not applied consistently across the globe, so appropriate general methods and calculation standards are required. In order to meet the needs of different stakeholders. researchers have tried to improve the existing energy efficiency metrics, but these are not enough to meet the new changes and challenges. In [111], several metrics were used to analyze the energy and cooling efficiency of small data centers in real-time. The authors came to the conclusion that new energy efficiency metrics need to be developed to evaluate the energy efficiency of data centers. For example, a new metric may be considered to be an extension of the work done by Cao et.al [112], which is on the basis of the two metrics of OEM(On-site Energy Matching) and OEF(Onsite Energy Fraction) [113]. They are shown as Equation (1) and Equation (2), respectively.

$$OEF = \frac{\int_{t_1}^{t_2} Min[G(t); L(t))] dt}{\int_{t_1}^{t_2} (L(t)) dt}; \quad 0 \le OEF \le 1.$$
(1)

$$OEM = \frac{\int_{t1}^{t2} Min[G(t)); L(t)] dt}{\int_{t1}^{t2} G(t) dt}; \quad 0 \le OEM \le 1.$$
(2)

Where G(t) represents on-site generated power (kW) at time t and L(t) represents load power (kW) at time t, and t1 and t2 represent the start and end of an energy efficiency evaluation in a given time period, respectively. These two metrics are used to evaluate mismatch problems between on-site renewable energy production(such as heat recovery in a data center, solar, and wind) and original configuration load. Yuventi et al. [114] proposed that PUE could only roughly convey the lowest possible energy consumption, but could not accurately convey the information over a period of time. Most of the new energy efficiency evaluation metrics proposed by researchers may appear in an integrated form in the future.

It is worth mentioning that, according to [89], PUE has been used as a mainstream approach for determining the energy efficiency of data centers in the global technology industry. Despite the PUE having some shortcomings, it is still an optimal energy efficiency evaluation method until a better alternative evaluation metric appears.

5.3. Multi-metric evaluation

Obviously, it is not objective enough that use a single metric to comprehensively evaluate the energy efficiency of a data center. The difference is that multi-metric evaluation can more accurately and comprehensively reflect the energy efficiency of the data center in some details because it comprehensively considers various dimensions and weighs different metrics. We assume that DCEE represents data center energy efficiency, which can be represented as Equation (3).

$$DCEE = \beta_1 * ITEU + \beta_2 * CUE + \beta_3 * WUE + \beta_4 * DCCSE + \beta_5 * ERE + \beta_6 * PUE.$$
(3)

Where ITEU denotes the IT equipment utilization, CUE is the carbon usage effectiveness, WUE is the water usage effectiveness, DCCSE is the data center cooling system efficiency, ERE is the energy reuse effectiveness, PUE is the power usage effectiveness, and $\beta_k (k = 1, ..., 6)$ represents the weight of the corresponding metric.

The selected metrics in Equation (1) are mainly from the perspective of green energy, as possible to achieve green data centers. Environmental protection plays an important role in global sustainable development, especially in the case of larger data centers. So establishing a new energy efficiency evaluation system is a common expectation of people.

The multi-metric energy efficiency evaluation process for data centers can be shown in Figure 3, which is designed by referring to the multi-objective decision-making methods in the sustainable energy decision [115]. The multi-metric energy efficiency evaluation process mainly includes four steps: determination of evaluation metrics, data processing, determination of metric weights, and combination of metrics and weights.

5.3.1. Determination of evaluation metrics

In multi-metric evaluation, we can't mix metrics at will because they usually have a limited scope of application. Not all metrics are suitable for multi-metric fusion. The selection of metrics follows several important rules:

- Metrics are numerical values that can be measured. Metrics measurement should not take up too much time or resources, and it should not disrupt normal operations.
- Metrics are oriented. The metrics aim to optimize the performance of the data center and have a traceable improvement direction.
- Metrics are independent. In the metric system, there is no overlap between metrics and no inclusion relationship.
- Metrics are practical and generalized. The measurement scale and granularity of metrics should meet the actual evaluation needs and can effectively measure multiple data centers, extensively.
- Metrics should be reliable and adaptable to changes in the data center's state.

5.3.2. Normalization of metrics

Different energy efficiency metrics have different dimensions and orders of magnitude. Therefore, when the level difference between metrics is large, if the original metric value is used for comprehensive evaluation, the role of higher value metrics in the a comprehensive evaluation will be highlighted, and the role of lower value metrics will be relatively weakened. To ensure the reliability of the evaluation results, it is necessary to normalize the original energy efficiency metric data. There are many methods for data normalization, the most popular are the min-max



Figure 3: Multi-metric energy efficiency evaluation process for data centers.

normalization and Z-score normalization. We assume that the original data set is defined as $X = \{x_1, x_2, ..., x_n\}$ and the processed result is $Y = \{y_1, y_2, ..., y_n\}$.

Min-max normalization: Min-max normalization is also called dispersion normalization in data processing. The transforming relationship is shown as Equation (4).

$$y_{i} = \frac{x_{i} - \min_{1 \le j \le n} \{x_{j}\}}{\max_{1 \le j \le n} \{x_{j}\} - \min_{1 \le j \le n} \{x_{j}\}}.$$
(4)

Where $\max_{1 \le j \le n} \{x_j\}$ is the maximum value in X, $\min_{1 \le j \le n} \{x_j\}$ is the minimum value in X and each y_i falls into the interval [0, 1]. This method retains the existing relationships in the original data and is the easiest way to eliminate the influence of dimensions and data range. When new data is added, the maximum and minimum values may change and need to be redefined. If a certain number in the value set is too large, the

normalized values are close to 0, and the difference between these values is small.

Z-score normalization: Z-score normalization is one of the most popular normalization methods, also known as standard deviation normalization. The results of data normalization are not all mapped into the interval [0, 1], which is different from the min-max normalization method. The normalization formula is shown as Equation (5).

$$y_i = \frac{x_i - \overline{x}}{s}.$$
 (5)

Where \overline{x} is the mean of the sample and $\overline{x} = \frac{1}{n} * \sum_{i=1}^{n} x_i$, s is the standard deviation of the sample and

$$s = \sqrt{\frac{1}{n-1} * \sum_{i=1}^{n} (x_i - \overline{x})^2}.$$

This method normalizes the data based on the mean and standard deviation of the original data. It is extremely applicable when the maximum and minimum values in the

original data set are unknown, or there is a value that exceeds the value range. After normalization, values of the variable fluctuate up and down around 0. Values larger than 0 indicate a higher value than the average, while values less than 0 indicate a lower value than the average.

5.3.3. Determination of metric weights

Weighting is a very important step in multi-index energy efficiency evaluation. The size of the weight indicates the importance of the corresponding metric, so different weight distributions will produce different values of energy efficiency evaluation, and reasonable weight can truly reflect the true situation of data center energy efficiency. There are three commonly used weighting methods, subjective weighting, objective weighting, and combining subjective and objective weighting.

Subjective weighting: Representative methods for setting subjective weights are analytic hierarchy process(AHP) [116] and Delphi [117]. In [118], authors studied the cloud computing data center adopting factors validity by fuzzy AHP successfully. Subjective empowerment may be selfcontradictory, and a consistency test needs to be added [119]. The Delphi method is an anonymous feedback inquiry method. Both methods are time-tested and are very mature and reliable weighting methods. They require experts in the industry to judge or score various metrics. Although they are subjective methods, multiple experts repeatedly perform this work to ensure the credibility of the weight setting. Moreover, experts can comprehensively consider the actual factors, so the weights will be more in line with actual situations. In the later stage, In order to form a unified standard, there is still needing an organization to call on data centers industry experts all over the world to communicate.

Objective weighting: Commonly used objective weighting methods usually include principal component analysis, entropy weight method, and mean square deviation method, multi-objective planning method. The objective weight is mainly coming from sample data, which eliminates the interference of subjective factors. However, objective weighting methods are also limited by sample data. When the sample data changes, the weight will also change, and the influence of outliers on the weight cannot be eliminated. In order to make the objective weighting effect more excellent and stable, the sample data must be large enough. At this stage, using data science methods, such as neural networks, linear regression, etc., to perform deep learning on samples to obtain objective weights is a relatively leading method. However, it takes huge manpower, material resources, and financial resources to obtain enough measurement data of energy efficiency metrics in cloud data centers. And the measurement methods of energy efficiency metrics are also controversial and have not yet reached an agreement in the industry. The premise of the intelligent weighting methods is difficult to achieve. So it is extremely difficult to use these methods to weigh multiple metrics, and it is still a vacancy at present. This shows that it is very important to

define a unified data centers construction standard. Which is not only conducive to industry specifications but also the establishment of a unified energy efficiency evaluation standard.

Combined weighting: Subjective weighting methods are closer to reality but are affected by the subjective preference of the decision-makers. Because the judgment of decisionmakers mainly depends on their knowledge or information, but subjectivity will affect the accuracy of the results. On the contrary, objective weighting methods obtain more scientific results by analyzing the original measurement data. But the weighting methods are too dependent on the sample. Although these two methods have a certain degree of information loss, the combination of subjective weight and objective weight can minimize information loss and make the weighted result as close to the actual result as possible. Subjective and objective weights are usually combined with multi-metric evaluation methods, such as the cloud model method, fuzzy comprehensive evaluation method, gray correlation evaluation method. The above objective weight methods can be found in [115]. Among them, the grey correlation method is more commonly used in energy systems [120, 121].

Data centers have complex structures and different scales, making it difficult to establish a unified universal energy efficiency evaluation metric system. The current multiindex evaluation of data centers energy efficiency still uses multiple metrics to classify the data center. The data center energy efficiency evaluation based on the integration of multiple metrics has not been adopted in reality. Moreover, the diversity of data center structures makes the multi-metric evaluation not suitable for the evaluation of data centers with a global concept. Multi-metric evaluation is suitable for evaluating the energy efficiency of a single data centers.

6. Summary

The commercial use of 5G and the rise of artificial intelligence had promoted the development of the digital economy. As one of the pillars of the digital economy, data centers have brought amazingly high energy consumption while growing rapidly. In order to promote sustainable development and improve the energy efficiency of data centers, researchers have developed a large number of energy efficiency evaluation metrics and some energy efficiency evaluation methods.

By evaluating the majority of the data center energy efficiency evaluation research experience, this paper described the energy efficiency evaluation techniques and three granular metrics. At the same time, we looked at the benefits and drawbacks of various evaluation methodologies and metrics, particularly the power usage efficiency (PUE). Then we addressed the current state of energy efficiency evaluation and the challenges it faces, as well as some potential improvements. PUE is the most often used evaluation metric today. To adapt to the fast-changing data center, we recommend

A Keview of Energy Efficiency Evaluation Technologies in Cloud Data Centers

boosting PUE performance. At the same time, we advocate for the creation of new metrics. The result of the multi-metric evaluation is more objective than the result of evaluating the data center by a single metric. We urge that data centers be evaluated using a multi-metric approach more often.

The data center industry, decision-makers, and academia have all made significant contributions to the evaluation of energy efficiency. Energy efficiency evaluation still has a long way to go, as evidenced by data centers' ever-increasing energy use. Academia should pay close attention to industry and integrate study findings from energy efficiency assessments into practice. Simultaneously, academics and industry should promote new technology aggressively in order to prevent its misuse. Renewable and alternative energy should be taken into consideration by data center operators. While boosting energy efficiency, the data center's carbon emissions are reduced. Studies have pointed out that, compared with large data centers, most of the energy consumption of data centers is consumed by small and medium data centers [23]. The majority of small and medium-sized data centers use outdated energy-saving technologies. Decision-makers should keep the number of small and medium data centers under control so that they are forced to use more energy efficient technologies.

The data center industry is responsible for developing energy efficiency evaluation standards. Energy efficiency has been enhanced to some extent thanks to the combined efforts of numerous stakeholders. However, data center operators, decision-makers, and academics must maintain their efforts in the face of rising energy usage. As the two largest power consumers in data centers, infrastructure (non-IT equipment) and IT facilities, both need to considerably increase energy efficiency. In the future, we may put more effort into energy efficiency modeling, energy-saving technologies, promotion of energy efficiency evaluation criteria, and the utilization of renewable resources to create more sustainable green data centers.

Acknowledgements

This work was supported in part by the National Natural Science Foundation of China under Grant No. 62172350, the Hunan Province Department of Education under Grant 21B0120, the Natural Science Foundation of Hunan under Grant No. 2021JJ40544, the National Natural Science Foundation of China under GrantNo. 62032020, No. 62172349 and No. 62076214.

References

- M. E. N., Aspen global change institute energy project. aspen global change institute., https://www.agci.org/project/energy- project, 2020.
- [2] W. S., Intels core i7-6700k skylake processor reviewed (2016).
- [3] Canalys, Global cloud services market reaches us\$42 billion in q1 2021, https://www.canalys.com/newsroom/ global-cloud-market-q121, 2021.
- [4] S. H. F, Ccid consulting ide statistical report., http://www.mtx.cn/#/, 2020.

- [5] Canalys, Global cloud infrastructure market q4 2019 and full year 2019. shanghai (china), bengaluru (india), singapore, reading (uk) and portland (us) c tuesday, 4 february 2020, https://www.canalys. com/newsroom/canalys-worldwide-cloud-infrastructure-Q4-2019-and-full-year-2019, 2020.
- [6] J. Yuventi, R. Mehdizadeh, A critical analysis of power usage effectiveness and its use in communicating data center energy consumption., Energy and Buildings 64 (2013) 90–94.
- [7] A. S. G. Andrae, T. Edler, On global electricity usage of communication technology: trends to 2030., Challenges 6 (2015) 117–157.
- [8] A. A., Total consumer power consumption forecast. nordic digital business summit, Challenges 10 (2017).
- [9] Z. R. Schodwell B, Erek K, Data center green performance measurement: State of the art and open research challenges., Americas Conference on Information Systems (2013).
- [10] H. Cheng, B. e. a. Liu, A survey of energy-saving technologies in cloud data centers, J. Supercomput. 77 (2021) 13385–13420.
- [11] C. G. Cloud., Index, cisco annual internet report (2018c2023), White Paper. (2020).
- [12] e. a. Masanet E, Shehabi A, Recalibrating global data center energyuse estimates, Science. 367 (2020) 984–986.
- [13] V. D. Reddy, B. Setz, G. e. a. Subrahmanya, Metrics for sustainable data centers, IEEE Transactions on Sustainable Computing. 2 (2017) 290–303.
- [14] P. M. Arregoces M, Data center fundamentals., Cisco Press. (2003).
- [15] G. Callou, J. Ferreira, P. e. a. Maciel, An integrated modeling approach to evaluate and optimize data center sustainability, dependability and cost., Energies. 7 (2014) 238–277.
- [16] R. Yadav, W. Z. et al., Managing overloaded hosts for energyefficiency in cloud data centers, Clust. Comput. 24 (2021) 2001– 2015.
- [17] S. Long, W. L. et al., A game-based approach for cost-aware task assignment with qos constraint in collaborative edge and cloud environments, IEEE Transactions on Parallel and Distributed Systems 32 (2021) 1629–1640.
- [18] S. Azizi, M. S. et al., Grvmp: A greedy randomized algorithm for virtual machine placement in cloud data centers, IEEE Systems Journal 15 (2021) 2571–2582.
- [19] S. N. Livieratos, S. P. et al., A new proposed energy baseline model for a data center as a tool for energy efficiency evaluation, International Journal of Power and Energy Research (2019).
- [20] A. F. Santos, P. D. Gaspar, H. J. L. de Souza, Evaluation of the heat and energy performance of a datacenter using a new efficiency index: Energy usage effectiveness design – eued., Brazilian Archives of Biology and Technology (2019).
- [21] R. A. Procaccianti G, Energy efficiency and power measurements: an industrial survey., ICT for Sustainability. Atlantis Press (2016).
- [22] X. Gong, Z. Z. et al., A review on evaluation metrics of thermal performance in data centers, Building and Environment 177 (2020) 106907.
- [23] T. L. Vasques, P. S. M. et al., A review on energy efficiency and demand response with focus on small and medium data centers, Energy Efficiency 12 (2018) 1399–1428.
- [24] K. Zhang, Y. Z. et al., Recent advancements on thermal management and evaluation for data centers, Applied Thermal Engineering 142 (2018) 215–231.
- [25] S. Singh, A. Swaroop, A. Kumar, Anamika, A survey on techniques to achive energy efficiency in cloud computing, 2016 International Conference on Computing, Communication and Automation (IC-CCA) (2016) 1281–1285.
- [26] C. Jin, X. Bai, C. Yang, Effects of airflow on the thermal environment and energy efficiency in raised-floor data centers: A review., The Science of the total environment 695 (2019) 133801.
- [27] J. Ni, X. Bai, A review of air conditioning energy performance in data centers, Renewable & Sustainable Energy Reviews 67 (2017) 625–640.
- [28] N. Horner, I. M. L. Azevedo, Power usage effectiveness in data centers: overloaded and underachieving, The Electricity Journal 29

(2016) 61–69.

- [29] Q. Zhang, Z. M. et al., A survey on data center cooling systems: Technology, power consumption modeling and control strategy optimization, J. Syst. Archit. 119 (2021) 102253.
- [30] W. Lin, F. S. et al., A taxonomy and survey of power models and power modeling for cloud servers, ACM Computing Surveys (CSUR) 53 (2020) 1 – 41.
- [31] A. Katal, S. Dahiya, T. Choudhury, Energy efficiency in cloud computing data center: a survey on hardware technologies, Cluster Computing (2021).
- [32] J. Aslan, K. Mayers, J. G. e. a. Koomey, Electricity intensity of internet data transmission: Untangling the estimates., Journal of Industrial Ecology 22 (2018) 785–798.
- [33] G. A. Brady, N. Kapur, J. L. e. a. Summers, A casestudy and critical assessment in calculating power usage effectivenessfor a data centre., Energy Conversion and Management 76 (2013) 155–161.
- [34] C. Belady, Green grid data center power efficiency metrics: Pue and dcie, 2008.
- [35] V. Avelar, D. Azevedo, A. e. a. French, Pue: acomprehensive examination of the metric., White paper (2012) 49.
- [36] W. Lin, H. W. et al., A cloud server energy consumption measurement system for heterogeneous cloud environments, Inf. Sci. 468 (2018) 47–62.
- [37] K. Shoukourian H, Forecasting power-efficiency related key performance indicators for modern data centers using lstms., Future Generation Computer Systems. 112 (2020) 362–382.
- [38] H. Shoukourian, T. W. et al., Using machine learning for data center cooling infrastructure efficiency prediction, 2017 IEEE International Parallel and Distributed Processing Symposium Workshops (IPDPSW) (2017) 954–963.
- [39] R. ul Islam, X. R. et al., Capacity management of hyperscale data centers using predictive modelling, Energies (2019).
- [40] J. C. Salinas-Hilburg, M. Z. et al., Fast energy estimation framework for long-running applications, Future Gener. Comput. Syst. 115 (2021) 20–33.
- [41] R. Calheiros, Cloudsim: a toolkit for modeling and simulation of cloud computing environments and evaluation of resource provisioning algorithms, Software: Practice and Experience 41 (2011).
- [42] D. Kliazovich, Greencloud: a packet-level simulator of energy-aware cloud computing data centers, The Journal of Supercomputing 62 (2010) 1263–1283.
- [43] S. S. Gill, S. T. et al., Thermosim: Deep learning based framework for modeling and simulation of thermal-aware resource management for cloud computing environments, J. Syst. Softw abs/2004.08131 (2020).
- [44] A. M. S. et al., Autoscalesim: A simulation toolkit for auto-scaling web applications in clouds, Simul. Model. Pract. Theory 108 (2021) 102245.
- [45] D. G. do Lago, R. C. da Silva et al., Sinergycloud: A simulator for evaluation of energy consumption in data centers and hybrid clouds, Simul. Model. Pract. Theory 110 (2021) 102329.
- [46] K. Alwasel, D. N. J. et al., Iotsim-osmosis: A framework for modeling and simulating iot applications over an edge-cloud continuum, J. Syst. Archit. 116 (2021) 101956.
- [47] P. C. Cañizares, A. N. et al., Mt-ea4cloud: A methodology for testing and optimising energy-aware cloud systems, J. Syst. Softw. 163 (2020) 110522.
- [48] W. Lin, S. Xu, L. He, J. Li, Multi-resource scheduling and power simulation for cloud computing, Inf. Sci. 397 (2017) 168–186.
- [49] J. E., Pue: The green grid metric for evaluating the energy efficiency in dc (data center). measurement method using the power demand., in: IEEE 33rd International Telecommunications Energy Conference (INTELEC), 2011, pp. 1–8.
- [50] R. N. Calheiros, R. Ranjan, A. e. a. Beloglazov, Cloudsim: a toolkit for modeling and simulation of cloud computing environments and evaluation of resource provisioning algorithms., Software: Practice and experience 41 (2011) 23–50.

- [51] K. Kliazovich D, Bouvry P, S. U., Greencloud: a packet-level simulator of energy-aware cloud computing data centers., The Journal of Supercomputing 62 (2012) 1263–1283.
- [52] B. D., A stochastic model to investigate data center performance and qos in iaas cloud computing systems., IEEE Transactions on Parallel and Distributed Systems 25 (2013) 560–569.
- [53] D. Shen, J. Luo, F. e. a. Dong, Stochastic modeling of dynamic rightsizing for energy-efficiency in cloud data centers., Future Generation Computer Systems 48 (2015) 82–95.
- [54] F. Sun, J. Yu, Improved energy performance evaluating and ranking approach for office buildings using simple-normalization, entropybased topsis and k-means method, Energy Reports 7 (2021) 1560– 1570.
- [55] M. Deymi-Dashtebayaz, M. Norani, Sustainability assessment and emergy analysis of employing the cchp system under two different scenarios in a data center, Renewable & Sustainable Energy Reviews 150 (2021) 111511.
- [56] H. Y. et al., Techno-economic assessment of building energy efficiency systems using behavioral change: A case study of an edgebased micro-moments solution, Journal of Cleaner Production 331 (2022) 129786.
- [57] T. Khan, W. T. et al., Workload forecasting and energy state estimation in cloud data centers: Ml-centric approach, Future Generation Computer Systems 128 (2022) 320–332.
- [58] X. Zhou, Y. M. et al., A dynamic energy benchmarking methodology on room level for energy performance evaluation, Journal of building engineering 42 (2021) 102837.
- [59] C. Peng, H. C. et al., A framework for evaluating energy security in china: Empirical analysis of forecasting and assessment based on energy consumption, Energy 234 (2021) 121314.
- [60] H. Zhao, B. L. et al., Economy-environment-energy performance evaluation of cchp microgrid system: A hybrid multi-criteria decision-making method, Energy 240 (2022) 122830.
- [61] J. Cho, J. Y. et al., Development of an energy evaluation and design tool for dedicated cooling systems of data centers: Sensing data center cooling energy efficiency, Energy and Buildings 96 (2015) 357–372.
- [62] Z. Kailong, X. C. et al., A novel performance assessment method of the carbon efficiency for iron ore sintering process, Journal of Process Control 106 (2021) 44–53.
- [63] Y. Xu, C. Y. et al., An anomaly detection and dynamic energy performance evaluation method for hvac systems based on data mining, Sustainable Energy Technologies and Assessments 44 (2021) 101092.
- [64] X. Lin, Z. T. et al., An energy performance assessment method for district heating substations based on energy disaggregation, Energy and Buildings (2021) 111615.
- [65] K. Al-Zoubi, G. A. Wainer, Fog and cloud collaboration to perform virtual simulation experiments, Simul. Model. Pract. Theory 101 (2020) 102032.
- [66] C. Peña-Monferrer, R. Sawko, V. Elisseev, Hpc-cloud native framework for concurrent simulation, analysis and visualization of cfd workflows, Future Gener. Comput. Syst. 123 (2021) 14–23.
- [67] M. Zakarya, L. Gillam, Modelling resource heterogeneities in cloud simulations and quantifying their accuracy, Simul. Model. Pract. Theory 94 (2019) 43–65.
- [68] M. S., Using bayesian networks for highly available cloud-based web applications., Journal of Reliable Intelligent Environments 1 (2015) 87–100.
- [69] H. Gonzalez A J, B. E., Hybrid cloud management to comply efficiently with sla availability guarantees., in: IEEE 12th International Symposium on Network Computing and Applications., 2013, pp. 127–134.
- [70] I. A. Nae V, Prodan R, Sla-based operations of massively multiplayer online games in clouds., Multimedia systems 20 (2014) 521–544.
- [71] P. B. Ferreira A M, Managing the complex data center environment: an integrated energy-aware framework., Computing 98 (2016) 709– 749.

- [72] T. Voort, V. ZAVREL, I. T. e. a. GALDIZ, Analysis of performance metrics for data center efficiency., REHVA Journal (2017).
- [73] C. R., European data centre infrastructure standards-en 50600 series., Nexans Cabling Solution (2016).
- [74] G. H., Data center handbook., John Wiley & Sons (2014).
- [75] J. Haas, J. Froedge, J. e. a. Pflueger, Usage and public reporting guidelines for the green grids infrastructure metrics (pue/dcie)., White Paper 22 (2009).
- [76] H. SegoL, A. Marquez, A. e. a. Rawson, Implementing the data center energy productivity metric in a high-performance computing data center., Design Technologies for Green and Sustainable Computing Systems, Springer 93-116 (2013).
- [77] J. M. Kaplan, W. Forrest, N. Kindler, Revolutionizing data center energy efficiency., Technical report, McKinsey & Company 93-116 (2008).
- [78] M. K. Patterson, S. W. Poole, C. e. a. Hsu, Tue, a new energyefficiency metric applied at ornls jaguar., in: International Supercomputing Conference, Springer, 2013, pp. 372–382.
- [79] G. T., Green grid metrics: Describing datacenter power efficiency., Green Grid Metrics WP. pdf (2007).
- [80] G. I., Promotion council, new data center energy efficiency evaluation index dppe (datacenter performance per energy) measurement guidelines., richtlinieCgreen it promotion council (gipc) (2012).
- [81] B. M., The green grid data center compute efficiency metric: Dcce., The Green Grid, White Paper 34 (2010).
- [82] Y. Taniguchi, K. Suganuma, T. e. a. Deguchi, Tandem equipment arranged architecture with exhaust heat reuse system for softwaredefined data center infrastructure., IEEE Transactions on Cloud Computing 5 (2015) 182–192.
- [83] T. Wilde, A. Auweter, M. K. e. a. Patterson, Dwpe, a new data center energy-efficiency metric bridging the gap between infrastructure and workload., in: International Conference on High Performance Computing & Simulation (HPCS), 2014, pp. 893–901.
- [84] P. Mahadevan, P. Sharma, S. e. a. Banerjee, A power benchmarking framework for network devices., in: International Conference on Research in Networking, Springer, 2009, pp. 795–808.
- [85] F. w. c. Subramaniam B, The green index: A metric for evaluating system-wide energy efficiency in hpc systems., in: IEEE 26th International Parallel and Distributed Processing Symposium Workshops & PhD Forum, 2012, pp. 1007–1013.
- [86] P. M. Pakbaznia E, Minimizing data center cooling and server power costs., in: Proceedings of the 2009 ACM/IEEE international symposium on Low power electronics and design, 2009, pp. 145– 150.
- [87] B. Tschudi, O. Vangeet, J. e. a. Cooley, Ere: A metric for measuring the benefit of reuse energy from a data center., White Paper 29 (2010) 82–83.
- [88] P. Mathew, S. Ganguly, S. e. a. Greenberg, Self-benchmarking guide for data centers: Metrics, benchmarks, actions., Technical Report, Lawrence Berkeley National Lab.(LBNL), Berkeley, CA (United States) (2009).
- [89] N. W. Schlitt D, Schomaker G, Gain more from pue: assessing data center infrastructure power adaptability., International Workshop on Energy Efficient Data Centers, Springer (2014) 152–166.
- [90] I. Rose, A. P. Wemhoff, A. S. Fleischer, The performance impact of integrating water storage into a chiller-less data center design, Journal of Thermal Science and Engineering Applications (2018).
- [91] M. E. Lei N, Statistical analysis for predicting location-specific data center pue and its improvement potential., Energy (2020) 117–556.
- [92] C. G. Cloud., Index, forecast and methodology, 2016-2021 white paper., Updated: February 1 (2018).
- [93] C. Fiandrino, D. Kliazovich, P. e. a. Bouvry, Performance and energy efficiency metrics for communication systems of cloud computing data centers., IEEE Transactions on Cloud Computing 5 (2017) 738– 750.
- [94] Council, Storage performance., Storage performance council. (2018).

- [95] D. Rydning, The digitization of the world from edge to core., Framingham: International Data Corporation. (2018).
- [96] G. G. organization, Harmonizing global metrics for data center energy efficiency global taskforce reaches agreement regarding data center productivity (2014) 1–29.
- [97] ISO, IEC, Iso/iec 30134-1:2016, information technology- data centres-key performance indicators -part 1: Overview and general requirements (2016) 1–18.
- [98] ISO, IEC, Iso/iec 30134-2:2016, information technology- data centres-key performance indicators -part 2: Power usage effectiveness(pue) (2016) 1–32.
- [99] ISO, IEC, Iso/iec 21836:2020, information technology -data centres -server energy effectiveness metric (2020) 1–78.
- [100] ISO, IEC, Iso/iec 23544:2021, information technology -data centres -application platform energy effectiveness(apee) (2021) 1–31.
- [101] CEN, CENELEC, E. et al., Review of standardisation activities energy management and environmental viability of data centres (2021) 1–20.
- [102] M. D. Avgerinou, P. Bertoldi, L. Castellazzi, Trends in data centre energy consumption under the european code of conduct for data centre energy efficiency, Energies 10 (2017) 1–18.
- [103] B. T., Global warming: Data centres to consume three times as much energy in next decade, experts warn., The Independent 23 (2016).
- [104] A. A. S., Projecting the chiaroscuro of the electricity use of communication and computing from 2018 to 2030., Researchgate net. (2019).
- [105] L. Belkhir, A. Elmeligi, Assessing ict global emissions footprint: Trends to 2040 & recommendations, Journal of Cleaner Production 177 (2018) 448–463.
- [106] ASHRAE, Data center energy standard: Ansi,ashrae standard., atlanta: Ashrae (2016) 1–39.
- [107] A. F. Santos, J. L. Heraldo, D. G. Pedro, Evaluation of the heat and energy performance of a datacenter by a new efficiency index: Energy usage effectiveness design-eued., in: The 25th IIR International Congress of Refrigeration(ICR 2019), volume 62, 2020, pp. 145–150.
- [108] D. e. a. Gilbert P, Ramakrishnan K, From data center metrics to data center analytics: How to unlock the full business value of dcim., CA technologies: white paper (2013).
- [109] T. Voort, V. e. a. Zavrel, Analysis of performance metrics for data center efficiency : should the power utilization effectiveness pue still be used as the main indicator? (part 2), 2017.
- [110] A. Zoie R C, Mihaela R D, S., An analysis of the power usage effectiveness metric in data centers., in: 5th International Symposium on Electrical and Electronics Engineering (ISEEE), IEEE, 2017, pp. 1–6.
- [111] M. G. Numata M, Sugiyama M, Barrier analysis for the deployment of renewable-based mini-grids in myanmar using the analytic hierarchy process (ahp)., Energies 13 (2020) 1400.
- [112] F. Huang, Y. Wu, J. Y. e. a. Wang, Study on comprehensive evaluation model for combined cooling heating and power system (cchp)., Journal of Engineering Thermophysics (2005).
- [113] J. J. Wang, Y. Y. Jing, C. F. Zhang, Weighting methodologies in multi-criteria evaluations of combined heat and power systems., International Journal of Energy Research 33 (2009) 1023–1039.
- [114] B. Lajevardi, K. R. Haapala, J. F. Junker, Real-time monitoring and evaluation of energy efficiency and thermal management of data centers., Journal of Manufacturing Systems 37 (2015) 511–516.
- [115] O. Gozc, B. Ozada, M. U. e. a. Carfi, Worldwide energy analysis of major free cooling methods for data centers., in: 16th IEEE Intersociety Conference on Thermal and Thermomechanical Phenomena in Electronic Systems (ITherm), 2017, pp. 968–976.
- [116] G. J., Machine learning applications for data center optimization. (2014).
- [117] T. Daim, J. Justice, M. e. a. Krampits, Data center metrics: An energy efficiency model for information technology managers., Management of Environmental Quality 20 (2009) 712–731.

A Review of Energy Efficiency Evaluation Technologies in Cloud Data Centers

- [118] J. J. Wang, Y. Y. Jing, C. F. e. a. Zhang, Review on multicriteria decision analysis aid in sustainable energy decisionmaking., Renewable and sustainable energy reviews 13 (2009) 2263–2278.
- [119] Z. F., The analytic hierarchy processa survey of the method and its applications., Interfaces 16 (1986) 96–108.
- [120] e. a. Linstone H. A, Turoff M, The delphi method., Addison-Wesley Reading, MA (1975).
- [121] N. Badie, A. R. C. Hussin, A. H. Lashkari, Cloud computing data center adoption factors validity by fuzzy ahp., International Journal of Computational Intelligence Systems 8 (2015) 854–873.